

Received March 4, 2020, accepted March 23, 2020, date of publication March 26, 2020, date of current version April 14, 2020. *Digital Object Identifier* 10.1109/ACCESS.2020.2983601

When Blockchain Meets Supply Chain: A Systematic Literature Review on Current Development and Potential Applications

SHUCHIH E. CHANG AND YICHIAN CHEN¹⁰

Graduate Institute of Technology Management, National Chung Hsing University, Taichung 40227, Taiwan Corresponding author: Yichian Chen (cycx1000@gmail.com)

This work was supported by the Ministry of Science and Technology, Taiwan, under Contract MOST-106-2221-E-005-053-MY3.

ABSTRACT This study aims to explore the current status, potential applications, and future directions of blockchain technology in supply chain management. A literature survey, along with an analytical review, of blockchain-based supply chain research was conducted to better understand the trajectory of related research and shed light on the benefits, issues, and challenges in the blockchain-supply-chain paradigm. A selected corpus comprising 106 review articles was analyzed to provide an overview of the use of blockchain and smart contracts in supply chain management. The diverse industrial applications of these technologies in various sectors have increasingly received attention by researchers, engineers, and practitioners. Four major issues: traceability and transparency, stakeholder involvement and collaboration, supply chain integration and digitalization, and common frameworks on blockchain-based platforms, are critical for future orientation. Traditional supply chain activities involve several intermediaries, trust, and performance issues. The potential of blockchain can be leveraged to disrupt supply chain operations for better performance, distributed governance, and process automation. This study contributes to the comprehension of blockchain applications in supply chain management and provides a blueprint for these applications from the perspective of literature analysis. Future efforts regarding technical adoption/diffusion, block-supply chain integration, and their social impacts were highlighted to enrich the research scope.

INDEX TERMS Blockchain, digital ledger, distributed ledger technology, logistics, shared ledger, smart contract, supply chain management, systematic literature review, value chain.

I. INTRODUCTION

In recent years, a proliferation of research, projects, and discussions regarding distributed ledger technology (DLT) has increasingly attracted the attention of researchers and practitioners. The reason lies in the characteristics that DLT may deliver promising disruption to the current model of trust, which has long created operational pain points of centralized systems. Traditional business operations rely heavily on a centralized authority or third parties, such as banks, to promote trust among participants [1], [2]. However, they have often been the targets of malicious attacks, malfunction and artificial alterations. The emergence of blockchain technology may bring about a rethinking of the design of business operations by virtue of its distributed and decentralized characteristics [3].

The associate editor coordinating the review of this manuscript and approving it for publication was Xiang Zhao¹⁰.

Blockchain, a DLT, refers to a consecutive list of timestamped records (usually digital transaction data) sequentially linked using cryptography. A peer-to-peer network of participating nodes contribute to the formation and validation of blockchain and manages distributed consensus by network majority. This makes blockchain an immutable, secure, and trustless model where transactions among parties are concerned. Blockchain technology has the potential to rebuild the way businesses conduct their operations [1], [4]. Several use cases have been investigated using the concept of blockchain, which could function as a distributed database without third parties. Several research endeavors regarding trade finance [5], medical record management [6], voting [7], and insurance industries [8] have improved the visibility of blockchain. Supply chain management (SCM) is one of these potential applications [9]-[13].

Growing efforts have been devoted to the study of blockchain technology and its applications across various

sectors [9]. The scope of this research ranges from technical discussion [14], [15], application feasibility [16], business operations [4], to even legal issues [17] and user acceptance [18]. Gathering information from 106 articles, including journal and conference papers published in a variety of sources during the period 2016 to early 2020, this study presents illustrative topics related to blockchain evolution in the recent decade. The major research methods adopted in these studies have also been reported to outline the relationship between the main topics and methodologies.

Among extant literature, research regarding the application of blockchain technology in supply chains is rather fragmented and diverse in topics [19], [20]. This phenomenon indicates a promising research interest but less-organized comprehension among the literature distribution. While researchers presented their early efforts regarding blockchain-based supply chain literature, there was a lack of qualified peer-reviewed articles that could be aggregated and analyzed in the current research's limited time frame. In [19], the first blockchain-related supply chain literature survey was conducted prior to January 2018 under scientific rigor with a peer-reviewed process. The authors also collected several blockchain trial pilots to demonstrate their contributions, and suggested second-round surveys to capture the rapid development of blockchain technology in the realm of SCM. Similarly, another study in the same-period [20] highlighted examples of blockchain-SCM integration while reporting the scarcity of related studies in leading journals and databases. In this study, the researchers highlighted the early adoption of blockchain in the electric power industry with a relatively mature understanding of blockchain-SCM integration. Both these papers reviewed only a few studies before 2018, i.e., 29 and 27 papers, respectively. These initial literature reviews are not adequate for providing a general overview, and a literature update is required due to the rapid proliferation of blockchain-based supply chain research from year 2018 to early 2020. Unlike the aforementioned review articles, this current work aims to provide a more holistic analysis of scenarios of blockchain use in supply chains by supplementing academic research for the past two years. Since blockchain with its unique features allows more streamlined business processes [21], [22], and enhanced transparency and trust among stakeholders, exploring its applications in supply chains is important for both academics and practitioners. We also realize the need for reporting the current status and potential of blockchain and its emerging applications across different industries in the SCM context. In this regard, we proposed certain questions to develop the logical connection between relevant extant articles and the potential for further research.

The research questions in this study are as follows:

RQ1: What are the main topics and subjects of interest in supply chain studies that utilize blockchain technology; how do they address its core issues; and how have these topics evolved over time? RQ2: What are the main research methodologies employed in blockchain-based supply chain literature and how are they related to the main topics?

RQ3: Which blockchain-supply-chain papers were most instrumental in driving the development of literature thus far?

To answer these questions, we adopted a systematic literature review (SLR) to gain insights regarding correlated issues. The aim of using SLR was to present a general overview of recent research by conducting a systematic analysis of extant literature. Thus, we outline the understanding regarding the research path based on our comprehension of related research interests and topic distributions. This paper contributes to the understanding of blockchain applications in the field of SCM by exploring various research topics and directions for future research. The literature analysis highlighted several potential research areas that may point out certain research gaps for future study. The remainder of this paper is organized as follows. Section II describes the basic knowledge regarding blockchain and smart contracts. Section III presents the methodology and procedural steps utilized in this review and Section IV reports findings from the analysis of literature. Section V identifies the main research issues, challenges, and implications, before the final conclusion is presented in Section VI.

II. PRELIMINARIES

A. BLOCKCHAIN

Nakamoto [23] proposed the concept of a decentralized digital currency, Bitcoin, supported by a decentralized payment system. Decentralization refers to an operating mechanism that allows peer-to-peer (P2P) exchange or transactions without centralized authorities. This disruptive innovation eliminates the heavy reliance on powerful third parties. Blockchain is the technology underpinning the Bitcoin cryptocurrency, which is a consecutive growing list of blocks, wherein each block records encrypted transactional data and may have further potential for other decentralization purposes [24]. The operating nodes in this kind of collaborative network have a duplicate record of transactional information, known as a "ledger." Inherently shared by participating nodes, DLT provides the opportunities for a trustless operating environment without traditional trusted authorities such as banks and clearing houses. P2P exchange facilitates trust building among participating nodes and the shared ledger is maintained by nodes in the network.

Computer nodes comply with an encrypted protocol to verify updated data in the shared ledger. This kind of distributed ledger system harvests the benefits of decentralized governance which may solve the issue of information exposure and accountability [25]. This inherent attribute favors the interactions between counterparties in the context of business operations [26]. Critical information could be maintained without checking the consistency of individual data and every single node possesses a duplicate of transactional data, thereby enhancing the transparency and visibility of business activities.

In a supply chain context, this kind of system and operational scheme may provide a better foundation of trust as well as benefits resulting from the absence of a centralized authority and intermediation [12]. Accordingly, blockchain could further be utilized to record the ownership of assets [27], permissions, and activity logs. This improves the traceability of information, cash, and process flows, and thus provides timely tracking of products and services. The different types of blockchain are public, private, and consortium (or federated) blockchains, each of which could be applied in certain scenarios to gain better advantages and for effectiveness.

B. SMART CONTRACTS

Smart contracts are programmable protocols that allow the execution of contract terms and agreements. The concept was first proposed by Szabo [28], who defined a smart contract as "a computerized transaction protocol that executes the terms of a contract." Thus, smart contracts can be deployed to a blockchain database and users can develop computer codes based on contractual clauses [29]. Contracts are executed when certain preset conditions are met. Smart contracts may outperform traditional contracts due to advantages such as allowing the mitigation of intermediaries among transacting counterparties and the facilitation of transaction flows without malicious alterations and tampering [30]. Smart contracts can be deployed on blockchain platforms in terms of scripts and stored with specific addresses for functional calls similar to those performed in other programmable computer languages [31]. They may facilitate data-driven interactions in the blockchain network and further allow applications to meet supply chain objectives.

Smart contracts are essential in the use of blockchain applications. A potential use case for smart contracts is in SCM. For example, a comprehensive paper-based process of administrative auditing may hinder the overall performance of a supply chain. Regulated conditions and agreements can be coded in smart contracts to avoid fraud, theft or other managerial risks. When smart contracts are deployed in the blockchain network, they are transferred to each connected node. Latest changes recorded in the local database may in turn trigger the conditions prescribed in computer codes to execute related process flows or notifications [32]. This is referred to as an event-driven mechanism, which can be facilitated without interference from a single entity. Typical use cases include conditioned payment transactions [33], [34], and asset and proprietary transfers [35].

With data updates and an event-driven mechanism, smart contracts may facilitate the manipulation of supply chain activities [36]. For example, commercial pilots, such as IBM and Maersk, recently announced their successful adoption of blockchain technology and smart contracts for shipment tracking and facilitating trade finance. Traditional paperbased procedures dealing with the collection and presentation of commercial documents could be significantly mitigated with the use of smart contracts [37]. This technology can reduce transaction costs and the number of involved intermediaries, and improve trade efficiency to achieve process automation.

Smart contracts on a blockchain-based platform can have applications in various fields; however, previous studies have reported certain unresolved issues with respect to technical and legal concerns [38]. For example, contract vulnerabilities with regard to transaction-ordering and timestamp dependence, mishandled exceptions, re-entrance, and callstack issues [39]. To pursue a wider spectrum of applications, more research endeavors must focus on solving validation/verification issues [40], enhancing security and privacy, and contract-based integration with other technologies [41]–[43]. Smart contracts may facilitate obligation execution and process automation among parties based on technical openness, however, its long-term development still requires a cross-disciplinary approach, combining technological, economic, and legitimation practices.

III. METHODS

To answer the research questions, we conducted an SLR to gain further insights into the relationship, integration, and development of blockchain technology in the realm of SCM. SLR is a kind of review genre that places strong emphasis on utilizing a series of standard processes to search, screen, analyze, and review literature, along with providing comments or suggestions. Since blockchain and its affiliated technology, i.e., smart contracts, have only emerged in the past decade, related works, technological standards, and innovative applications are still in their nascent stages. We selected the latest four years, 2016~2019 (extended to early 2020), as the time span of this literature survey to capture the most updated usage of these technologies in SCM. This study followed previous research guidelines to construct a robust and replicable study [44]–[46]. Additionally, to explore and analyze as many extant studies related to blockchain and supply chain as possible, we searched the IEEE Xplore, ACM, Google Scholar, and other major digital libraries from various academic publishers, using the following search string shown in Fig. 1. In this study, literature search and screening involved the following steps (Fig. 1).

A. STEP 1: INCLUSION/EXCLUSION CRITERIA

Article selection was conducted by setting inclusion and exclusion criteria prior to analysis to extract highly-relevant literature. In this research, we focused on keywords related to the emerging blockchain technology and its extended concept, smart contracts, in the field of SCM to identify relevant articles. Considering the essence of blockchain, more general synonyms or terminologies, such as DLT and "decentralized ledger system" (DLS), were included. After employing these procedures, we obtained a total of 433 articles. Meanwhile, exclusion criteria were set to exclude ambiguous articles not highly correlated to the key topics. In this research, we included some conference articles and proceedings to extend



FIGURE 1. Procedural steps of the selection protocol.

the scope of literature. However, preliminary studies in the form of technical reports, comments, editorial, and consulting papers, were excluded to ensure quality, consistency, and academic rigor.

B. STEP 2: TITLE AND ABSTRACT EXAMINATION

Based on the results of keywords identification and screening procedures, we examined the title and abstract of each selected article by each author. To assure quality and academic rigor, content of abstracts with less focus on major topics or with less concrete findings/contributions to the body of knowledge were removed. We double-checked our process to maintain coherence in article selection. In this stage, we excluded 294 papers that focused mainly on the applications of other emerging technologies such as the internet of things (IoT), cloud computing, industry 4.0, machine to machine (M2M), and radio frequency identification (RFID), etc.

C. STEP 3: FULL TEXT READING AND FINER SELECTION

To better align the research questions with the contents of the selected papers, we conducted a full text reading and perform the procedure for finer selection of candidate papers. We removed 33 papers from the selection as they were less relevant to our main focus. In this final step, we obtained 106 papers for this study. Fig. 1 illustrates the procedural steps for paper selection and presents the selection criteria or main actions conducted to filter for the most related articles.

Additionally, to better comprehend of the selected corpus, we conducted a preliminary descriptive analysis to determine answers for the research questions. We classified the selected papers according to their topics and research methods to respond to RQ1 and RQ2. We also conducted a citation analysis to provide evidence for RQ3. A number of frequently-cited supply chain articles based on blockchain and smart contract technology were examined to obtain a better understanding



FIGURE 2. Paper distribution by year of publication.

of the research interest and its applications. The following sections provide explanations for the literature classification and related analyses.

IV. RESULT

A. DESCRIPTIVE ANALYSIS OF THE CORPUS

Fig. 2 displays the distribution of the sources of the selected review articles and illustrates the proliferation of blockchain research in the field of SCM in the last 3 years. This indicates the potential and opportunities for this emerging technology in various facets of supply-chain-related issues. Blockchain was coined and introduced around 2008, from then on, research has focused on cryptocurrencies, e.g., Bitcoin, along with their security and privacy issues. The development and evolution of such technologies have moved from their original financial applications to other areas in the following years. During the period $2008 \sim 2015$, due to the technical essence of blockchain, related publications were scattered in the form of technical forums, consulting reports, news reviews, or comments. Since 2016, blockchain and its applications have drawn the attention of engineers, scholars, and practitioners. For example, the special topic conference entitled "The 2018 international conference on blockchain" focused on the discussion of applications of blockchain technology and smart contracts in various sectors, including supply chains.

Assorted but fragmented peer-reviewed journals contributed around 89% of the selected articles, while IEEE, ACM and other topic-related conferences accounted for the rest. Fig. 3 presents the popular journals for blockchain-related publications, which include *IEEE Access*, *International Journal of Information Management, Future Generation Computer Systems, Supply Chain Management: An International Journal, International Journal of Production Research, Sustainability, Computers & Industrial Engineering*, etc. Originally designed for solving the double spending issue of digital currency, blockchain applications

TABLE 1. Article topics classification.

Topics	Description	Papers		
General Influence / Overview	Influence on SCM, collaborative organizations and discussion on blockchain feasibility / applicability.	1, 19, 20, 26, 47, 63, 68, 73, 97, 98, 100, 107, 112, 116, 117, 131, 132, 135		
Distributed Ledger Technology & Smart Contract	These topics include the discussion, application and evaluation of distributed ledger technology and smart contract.	30, 36, 82, 89, 91, 93, 95, 102, 111, 116		
Business Process Management and Quality Management	This category emphasizes the implementation, monitoring and execution of business process in real practice.	3, 16, 21, 22, 32, 37, 40, 50, 74		
Ownership Management	Focus mainly on identity, asset or ownership transfer during business transactions.	27, 67		
Information Sharing	This area includes how data, such as shipping, accounting or financial information is shared in business operations.	16, 37, 62, 74, 111, 113, 114, 116, 126		
Security and Privacy	This topic includes issues on counterfeit prevention, tamper-proof and privacy-keeping mechanism.	55, 62, 87, 90, 91, 92, 114, 116		
Business Model and Operation	Business model describes the rationale of how organization create, deliver, and capture value, in economic, social, cultural contexts (Wikipedia). Business operation deals with work flow and coordination of processes.	1, 4, 73, 78, 105, 107, 117		
Acceptance and Adoption	This issue deals with the extent of the act taken or received by using blockchain technology in supply chain activities.	10, 18, 77, 78, 79, 80, 128, 129, 133 134		
Supply Chain Integration	Supply chain integration refers to the alignment and coordination within a supply chain. Focus is mainly stressed on the shared information system.	20, 115		
Purchasing and Supply	The process business participants go through to fulfill their demand on raw material, product and even service.	122		
Manufacturing	A process to produce merchandise for use or sale.	53, 54, 64, 66		
Physical Distribution and Logistics	Major concerns on the practical involvement of commodity transportation, shipping and distribution.	48, 52, 65, 72, 73, 74, 81, 82, 93, 101, 104, 105, 116, 123, 128, 130		
Transaction Processing	This topic deals with the operation of blockchain-based transaction processing.	33, 96, 123		
Financing, Accounting and Payment	A distributed and disintermediated method to process accounting, financing and payment.	33, 34, 37, 66, 130		
Governance	Business governance depicts a working mechanism which comprise of policies, methods and business processes for an organization's operation.	95, 96		
Credit Evaluation	A process businesses or lenders usually undertake when evaluating a request for credit.	94		
Blockchain with Other Technology	Internet of Things (IoTs) integration, Radio Frequency Identification (RFID), Industry 4.0, Machine to Machine (M2M), Near Field Communication (NFC)	55, 60, 68, 69, 70, 71, 84, 85, 92, 97, 104, 122, 127		
Traceability & Transparency	Traceability refers to the extent business participants could monitor or get full control of entity status. Transparency is about to the visibility of information and process flow.	16,18,30,47,49,51,50,52,54,57,58,59,67, 70,72,82,84,85,87,88,89,94,96,103,106, 110,116,123,124,125,126,127,132		
Agricultural & Food Applications	This topic deals with blockchain application in the agricultural and food industry.	50, 51, 59, 69, 84, 85, 86, 87, 88, 94, 106, 109, 110		
Other Industry Applications	Blockchain applications in drug, chemical, medical, construction, or other industry. 49,55,56,57,60,61,62,63,6 100,108,120			
Tactical Planning / Risk Analysis	Business strategy with tactical thinking. Consideration, classification and 75, 76, 118, 120 volu			
N. E. I.		· · · · · · · · · · · · · · · · · · ·		

Note: Each paper may refer to more than one topic. Paper number refers to that in Reference section.



FIGURE 3. Journal/conference publications and number of papers published.

were first used mainly in the financial sectors rather than others, such as supply chain, medical records, and information systems. Additionally, we discovered a growing demand for papers in 2018 and expect double or triple the amount of publications in 2019 and 2020. The growing numbers of conferences and special sections regarding blockchainrelated supply chain publications in recent years have resulted in an increasing research interest in this field. Due to the proliferation of blockchain-related conferences held in recent years, we excluded conference papers from the year 2019, but selected a few frequently-cited conference papers for the literature analysis.

B. CLASSIFICATION OF SELECTED PAPERS

To answer RQ1 and determine corresponding research methods for RQ2, we classified the selected papers in terms of supply-chain-centric topics to identify the research foci and examine the potential influence of and emphasis on blockchain in supply chain applications [47]. The classification of selected paper is presented in Table 1 with a brief description of related topics. A considerable number of papers covered traceability/transparency issues and the influence of blockchain technology in the supply chain context. Similar efforts highlighted the potential of blockchain-based supply chain applications in diverse sectors, prototype system

TABLE 2. Topic classification in various industries.

Topic assorted by application sectors	Papers
Physical Distribution and Logistics	48, 52, 65, 72, 73, 74, 81, 82, 93, 101, 104, 105, 116, 123, 128, 130
Agricultural & Food Applications	18, 50, 51, 52, 59, 69, 84, 85, 86, 87, 88, 109, 110
Manufacturing	53, 54, 64, 66
Drug and Pharmaceuticals	55, 56, 57, 96
Airline	63,64
Chemical industry	60
Construction	61
Other industry application	49, 67, 80, 100, 108, 120

Note: Paper number refers to that in Reference section.

design [48]–[50], and a combination of other emerging technologies. We noticed that each paper may have referred to more than one topic to better respond to the scope of the selected papers.

For better objectivity, we conducted a two-stage review for paper classification. We independently classified the selected papers into a grouping list. Next, we performed a cross examination to extract common opinions regarding research topics, finely adjusting for topic headings after the discussion. Subsequently, we conducted a random check using the previous classification to ensure less disagreement and to present more stable final decisions.

Table 2 presents potential blockchain applications across different industries showing that early applications were mainly in the fields such as food provenance [51], physical distribution and logistics [52], manufacturing [53], [54], and drug/pharmaceuticals supplies [55]–[57]. Traceability [58] and transparency [59] issues are increasingly drawing more attention, and this phenomenon explains the penetration point where blockchain could be applied in SCM. We also found early evidence that these applications may cover various sectors such as the chemical [60], construction [61], [62], airline [63]–[65], and manufacturing [66], [67].

The most addressed topics were divided into four categories: (1) traceability and transparency, (2) general influence/overview, (3) blockchain applications in agriculture or other industries, and (4) physical distribution and logistics (Fig. 4). The results of our topical discussion were reasonable since the major focus of the research interest and proposed solutions began with the pursuit of major supply chain objectives. We also found complementary interactions with emerging technologies such as IoT [68], RFID [69], [70], [71], NFC, etc., which are considered critical to streamlining supply chain flows/activities and real-time response.

C. ANALYSIS OF THE RELEVANCE OF THE TOPICS OVER TIME

Blockchain-related studies can be chronically divided into four-year intervals to examine the evolution in research topics. In this sense, we could further illustrate the relevance of research topics through an evolutional timeline. Before 2016, blockchain discussions mainly focused on Bitcoin or other digital currencies. Therefore, technical applications of blockchain in SCM were not significantly investigated. Fig. 5 illustrates topic relevance over time with the gray area denoting new topics. In the first period (year 2016), only few papers were published, and as expected, supply chain traceability and transparency were the first topics discussed. A few applications focusing on agricultural products were presented as templates. In the second stage (year 2017), it was apparent that growing attention had been granted to the discussion of blockchain integration with other emerging technologies, general influence of blockchain, the essence/design of the ledger system and smart contracts, and security/privacy issues. Simultaneously, the focus on the previous stage's research topics increased both in amount and in depth. More publications that explored traceability and transparency were presented and there was also a growing interest in practical applications among various industries.



FIGURE 4. Papers classified by topic.

In 2018, based on the extension of topics covered in the previous stage, academic efforts were increasingly devoted to blockchain security and privacy, distributed ledger technologies, and smart contracts. We also learned that blockchain applications in various sectors, e.g., manufacturing, were drawing more attention from technical engineers, practitioners, and scholars. Additionally, more papers in the four-year period investigated blockchain innovation from the perspective of business process reengineering (BPR). In 2019, an explosive interest in various topics, such as blockchain-enabled physical distribution and logistics [72], business process management, information sharing, business operations [73], [74], and risk analysis [75], [76], dramatically emerged. Specifically, empirical studies focusing on blockchain acceptance/adoption [77], [78] appeared to further provide evidence for previous technological forecasts, business models, and proto-concepts [79].

D. MAIN RESEARCH METHODOLOGIES EMPLOYED AND THE RELATIONSHIP WITH MAIN TOPICS

We followed a similar method of topic grouping to classify research methodologies. Two classification lists were generated and then discussed to diminish disagreement in the final version (Fig. 6).

Since blockchain is an emerging information technology, few studies utilized quantitative methods. Instead, we found more descriptive papers using case studies, proof of concept (POC), theory building, and literature reviews, etc. It should also be noted that there was a number of papers that explored blockchain's technical issue by system design, code/algorithm analysis, and simulation. Table 3 summarizes the connection between research topics and methods. We observed that various methods were adopted when discussing blockchain applications. Since blockchain is in its early development phase, quantitative methods were rarely used in prior research. Few studies used questionnaires and interviews [80] to collect data for qualitative rather than quantitative analysis [18], [81].

Qualitative methods were widely used to describe, predict, and propose blockchain opportunities in the supply chain context while quantitative methods were utilized to evaluate or analyze the performance of various blockchainbased systems. Case studies and comparative analysis have frequently been used to explore the potential influence of blockchain-based supply chains, security and privacy, and



2016		2017		2018		2019 ~	
Topics	Number of papers						
Traceability & Transparency	2	Traceability & Transparency	4	Traceability & Transparency	9	Traceability & Transparency	19
Business Process Management and Quality Management	1	Blockchain with Other Technology	4	General Influence / Overview	7	General Influence / Overview	10
Manufacturing	1	Security and Privacy	3	Other Industry Applications	4	Physical Distribution and Logistics	10
Blockchain with Other Technology	1	Physical Distribution and Logistics	3	Blockchain with Other Technology	3	Agricultural & Food Applications	9
Agricultural & Food Applications	1	Distributed Ledger Technology & Smart Contract	2	Agricultural & Food Applications	3	Acceptance / Adoption	8
		General Influence / Overview	2	Security and Privacy	3	Other Industry Applications	6
		Other Industry Applications	2	Manufacturing	3	Business Process Management and Quality Management	6
		Information Sharing	2	Distributed Ledger Technology & Smart Contract	3	Information Sharing	6
		Business Process Management and Quality Management	1	Information Sharing	2	Business Model and Operation	6
		Business Model and Operation	1	Acceptance / Adoption	2	Blockchain with Other Technology	5
		Financing, Accounting, and Payment	1	Financing, Accounting, and Payment	2	Tactical Planning / Risk Analysis	5
		Ownership Management	1	Physical Distribution and Logistics	2	Distributed Ledger Technology & Smart Contract	4
		Agricultural & Food Applications	1	Transaction Processing	2	Financing, Accounting, and Payment	2
		Transaction Processing	1	Business Process Management and Quality Management	1	Ownership Management	1
		Supply Chain Integration	1	Supply Chain Integration		Manufacturing	1
		Governance		Governance		Purchasing and Supply	
				Credit Evaluation		ar an ann an Granne an Abra	<u> </u>

FIGURE 5. Topic relevance over time. Note: Grey background refers to new topics compared to previously listed ones.



FIGURE 6. Research methods adopted in the corpus.

applications in various sectors. Several of the selected papers, from a technical-oriented perspective, utilized a blockchainbased system design and proposed conceptual frameworks for future system implementation. Some papers included coding segments to enhance understanding, and experimental simulations to assess the performance of blockchain-based systems.

Some authors developed questionnaires and conducted interviews to assess the influence, acceptance, and performance of improvements in supply chain traceability/ transparency due to blockchain. Very few papers conducted literature reviews due to the lack of adequate relevant literature. Additionally, some papers focused on the validation and evaluation of prototypes while few paid attention to blockchain-supply-chain theory building. Thus, most research studies adopted system design, case studies, or descriptive analysis to verify the effectiveness and potential of innovative blockchain-supply-chain projects/systems and associated conceptual frameworks.

E. CITATION ANALYSIS

Table 4 summarizes 12 frequently-cited documents (10 journal articles and two conference papers) in the corpus. To answer RQ3, we selected several typically-cited papers, categorized them, and provided complementary research findings for them. This selected list offers an outline of related topics about how blockchain may be applied in supply

Methods	Architecture Building & System Design	Case Study	Narrative Analysis Inductive Reasoning	Algorithm & Protocol	Survey & Interview	Theory/Model Building	Literature Review	SEM	Simulation Experiment	Others
General Influence / Overview	68	47,63,73,97	1,100,119,135			117,119,131,132	19,20,98,107			26,112,132
DLT & SC	36,82,89,91,102,116	30	95	111			93			
Business Process Management and Quality Management	16,21,22,37,50,74		40	32		32	3			
Ownership Management	67			27						
Information Sharing	16,37,74113,114,116	114		62,111					126	
Security and Privacy	91,114,116	55	90,92	62						
Business Model and Operation	105	73	1			4,117	107	78		
Acceptance and Adoption		80,128	129		10,18 80,133	77,128	80,134	77,78, 79,134		
Supply Chain Integration		115					20			
Purchasing and Supply		122								
Manufacturing		64,66		53		54				
Physical Distribution and Logistics	74,82,104,105,116,123	48,65,73,128,130	101		52,81	128	72,101		93	
Transaction Processing	96,123			33						
Financing, Accounting, and Payment	34,37	130	66	33						
Governance	96		95							
Credit Evaluation	94									
Blockchain with Other Technology	68,69,70,84,85,104,127	55,97,122	60,92	71						
Traceability and Transparency	16,49,50,51,52,57,70,82, 84,85,87,89,94,96,103, 116,123,124,125,127	30,47,52, 58,124	110	67	18	54,132	59,72,106		57,70,126	88,132
Agricultural & Food Applications	50,51,69,84,85,86,87,94		109,110				59,106			88
Other Industry Applications	49,56,57,67,96,108	55,63,64,80	60,100,120	61,62,108					108	
Tactical Planning / Risk Analysis	76	118	75,120			75,118			75	65

TABLE 3. Topics and methods used in the corpus.

Note: Paper number refers to that in Reference section

chains. Earlier-published papers may have been cited more frequently compared to more recent publications.

Early research focused on the examination and prediction of blockchain-enabled improvements on supply chain traceability and transparency [82], [83]. For example, Tian [84], [85] developed frameworks combining blockchain with RFID and the IoT. These papers serve as references for other applications in various industries, especially the agricultural sector [86], [87], [88]. Later, research interests moved toward the discussion of blockchain system design [89], security and privacy issues [90]–[93], credit evaluation [94], and governance [95], [96]. Evidence from pilot studies has urged scholars to examine blockchain's role in meeting SCM objectives [97]. Researchers have asserted that areas, such as extended visibility and traceability, supply chain digitalization and disintermediation, improved data security and smart contracts, are worthy of further study [19]. We suggest that a second-run SLR must be conducted to include more emerging research efforts regarding the various supply chain applications of blockchain.

V. DISCUSSION

Blockchain, with its unique characteristics, has received growing attention from engineers, researchers, and practitioners in the last decade. Through a series of explorations and implementations of several pilot projects in various sectors, blockchain technology has found practical applications from both academic and practical perspectives [98]–[101]. To answer the research questions in this paper as well as provide a better understanding of future blockchain developments, we attempted to bridge the gap in extant knowledge regarding blockchain applications and future research work with the findings from our literature review. The major research issues for the achievement of supply chain objectives are elaborated and discussed as follows.

A. TRACEABILITY AND TRANSPARENCY

Traceability and transparency have long been critical issues in supply chain activities. Typical pain points include critical intermediaries, process hand-offs, over-centralized business operations, etc. Blockchain, as a distributed shared ledger technology, may help increase traceability and extend supply chain visibility by its consensus mechanism and shared ledger. Every single node participating on a blockchain platform collaboratively maintains and validates transaction records in the common ledger. The major players, including suppliers, manufacturers, shippers, distributors, and customers, have duplicated transactional records and access permission to monitor the progress of process flows [102]. These inherent characteristics greatly reduce the need for trusted centralized authorities, allowing for immutable transactions and improved efficiency in supply chain activities.

TABLE 4.	List of	frequently	cited	documents in	ı the	corpus.
----------	---------	------------	-------	--------------	-------	---------

Paper	Highlights	Applied Scenarios	Key findings
[97]	Influence on SCM objectives	Global supply chains	Multiple case studies are conducted to address blockchain capabilities for achieving key supply chain management objectives.
	A animultural summity also in avatam	A cuiculture in ductors	A dual-chain ("user information chain" and "transaction chain") agricultural business resource public block chain is proposed
[80]	Agricultural supply chain system	Agriculture industry	to provide security guarantee mechanism and improve the utilization of business resources.
			This paper evaluates blockchain's roles in strengthening supply chain cybersecurity related to IoT devices and protecting
[92]	Security and Privacy	Global supply chains	privacy. Policy implications are suggested on establishment of public-private partnership and governmental provision of legal
			clarity on smart contracts.
140	Analitantum Investore denim	Manufastarian industra	A proposed blockchain architecture was developed to facilitate a high security, scalability and a well-structured cloud system
[00]	Architectural system design	Manufacturing industry	The design can bring more advantages to cloud manufacturing than the security and scalability.
		Development and second	Blockchain technology from the purchasing and supply management perspective may be envisioned to allow the flexible
[122]	Development of system design principles	Purchasing and supply	interconnectivity of autonomous purchasing systems and machine-to-machine interaction via smart contracts. Blockchain is
		management	viewed as a procurement value driver to product information records which is useful for supplier capability assessment.
			Gcoin-based blockchain as the base of the data flow of drugs is proposed to create transparent drug transaction data. The
[96]	Governance of blockchain-based system	Medical industry	regulation model of the drug supply chain could be altered from the inspection and examination only model to the surveillance
			net model. The proposed model is able to improve distributed governance and prevent drug counterfeits.
			Consumers are overwhelmed by the amount and complexity of certification labels. As a transparency and traceability system,
[18]	Blockchain acceptance	Food provenance	blockchain technology implementation appears to have significant positive influences on consumers' purchasing decisions,
			mediated by consumers' quality perceptions.
			A food supply chain traceability system for real-time food tracing based on HACCP (Hazard Analysis and Critical Control
[85]	Traceability with IoT	Agri-food industry	Points), blockchain and Internet of things, could provide an information platform for all the supply chain members with
			openness, transparency, neutrality, reliability and security.
			A RFID and blockchain combined system can realize the traceability with trusted information in the entire agri-food supply
[84]	Traceability with RFID	Agri-food industry	chain, which would effectively guarantee the food safety, by gathering, transferring and sharing the authentic data of agri-food
			in production, processing, warehousing, distribution and selling links.
			A systematic literature review in supply chain context shows trust as a key driver for blockchain adoption. The value lies in
[19]	Comprehensive literature review	Global supply chains	four areas: extended visibility and traceability, supply chain digitalization and disintermediation, improved data security and
			smart contracts.
			Empirical study using supply chain and network theory, as well as the technology acceptance models (TAMs), is conducted to
[79]	Adoption drivers	Global supply chains	provide evidence and report on blockchain adoption behavior at the individual level. The developed model investigated main
			drivers and reported distinct adoption behaviors between Indian- and USA-based professionals.
			A multiple case study is conducted to develop an explanatory model for the interaction of actors in a blockchain-based
[73]	Business operation	Global supply chains	operational supply chain. The study analyzed the impact of blockchain on replacing operational intermediaries and found
			certain blockchain's impacts on business models for logistics industry.

From the results we reviewed, we can note that nascent blockchain-based supply chain studies focus on the discussion of its potential to improve traceability and transparency [103]. Industrial pilots such as Provenance, Walmart, and Everledger have illustrated the capability to improve the visibility of physical movement, asset transfer, and quality assurance. The integration of blockchain and emerging technologies, such as the IoT and advanced sensing technology, will enable improved real-time monitoring of logistics activities [104]. By introducing better disclosure of supply chain activities and improving accountability across supply chain players, blockchain may mitigate disputes among businesses. With its immutable attributes, the distributed ledger system also reduces the need for reconciliation among traditional siloed databases. In this sense, stakeholders may capture value and alleviate risks when conducting transactions.

Moreover, smart contracts may facilitate process automation and provide better real-time monitoring of product/service visibility [105]. The design of smart contracts aims to achieve the functional demands of supply chain stakeholders. While trust is often weak among unfamiliar trading parties, blockchain may leverage its advantages to create value in supply chain activities. A blockchain ledger system may also provide authenticity verification of property or documents via digital signatures or other encrypted measures, which increases security and averts counterfeit [27], [71]. Enterprises may benefit from the reduced processing of paper-based documents as well save the considerable costs involved in tracking and obtaining proof of information authenticity. Accordingly, an improved supply chain working scheme with better traceability and transparency, less intermediation, and process automation is attainable when making blockchain transactions with counterparties. Moreover, blockchain's distributed governance and transparency attributes may also provide solutions to malicious alterations in a centralized system.

B. STAKEHOLDER INVOLVEMENT AND COLLABORATION

Joint participation of supply chain actors can enhance the overall performance of a blockchain ecosystem [106], [107]. From a systemic perspective, blockchain may leverage its potential through a large-scale collaboration of stakeholders as supply chain friction mainly stems from dispersed disconnection among parties [108]. However, blockchain may not be a panacea as certain challenges and barriers still remain in the further adoption of blockchain technology [109]–[111]. First, the willingness of supply chain stakeholders dominates the level of employment as blockchain-based architecture is still in its nascent stages. Some unsolved technical issues

include transaction throughput, scalability, security and privacy, block capacity, and power consumption issues. Supply chain participants are accustomed to the trust mechanism supported by centralized authorities and have less confidence in the effectiveness of a distributed trustless paradigm. Cultural mindset and extant business processes need to be changed before blockchain adoption.

Second, activities for business process redesign and migration need to be well planned. The successful establishment of a blockchain-based operating environment requires coordination among the legacy databases of various parties [75], [112]. Firms have to choose appropriate types of blockchain applications and scholars have reported several judging criteria for reference. For example, a private blockchain is suitable for an intra-organizational context as it has a higher level of centralization and security. Public blockchains are open to public participation without permission while a consortium blockchain provides mid-level of decentralization. Researchers have addressed the importance of establishing a governance model, considering blockchain configuration and operations among stakeholders. Additionally, information sharing and system interoperability are key concerns for blockchain architectural design [113], [114].

C. SUPPLY CHAIN INTEGRATION AND DIGITALIZATION

Blockchain has the technical capability to play a critical role in information sharing [115], [116], value/property transfer, and trust provision [108]. The way in which supply chain partners achieve strategic and business objectives using blockchain technology relies on the integration of business processes with the type of blockchain employed. From a value-creation perspective, blockchain enables the incorporation of a new business model [117] with disruptive innovation [118] and different levels of disintermediation, which allows for better transparency, security, privacy, and becomes a driving force of cooperation among untrusted parties. Blockchain-based financial applications grant a dramatic reduction on transaction costs with re-engineered processes since a synchronized ledger may significantly enhance the level of overall efficiency [119]. For example, researchers have suggested a blockchain-based letter of credit process, which may greatly improve efficiency in global trade context [16], [37].

Digitalization of physical properties and intelligence rights may be the next stage of development in blockchain application [19]. Digitalization allows the transfer of value across a distributed ledger system due to blockchain's capability for maintaining data integrity [120], free from malicious tampering and cyber-attacks [121]. Blockchain networks could also facilitate the transmission of digitalized stakes [122] and provide validation of document authenticity, which can mitigate transactional disputes. Moreover, blockchain and smart contracts facilitate process flows and improve the efficiency of payment transfers and settlements. Smart contracts, serving as digitalized protocols, could enable the tracking of supply chain events and document flows, which can streamline business operations and facilitate the execution of business logic. Therefore, supply chain digitalization is expected to promote stakeholder involvement in achieving common objectives.

D. COMMON FRAMEWORKS AND BLOCKCHAIN-BASED PLATFORMS

In a blockchain-based platform, smart contracts play a crucial role in connecting business logic and process execution in supply chain activities. For example, event-driven mechanisms are widely adopted when designing smart contract communications [37], [123]. Programmable codes are utilized to allow triggers when supply chain activities change states. For example, smart contracts trigger corresponding notifications to supply chain stakeholders when certain preset conditions are met, such as goods arrival, payment received, or shipping documents approved. Trial pilots with a variety of blockchain platforms have been devoting efforts to realize the better performance of supply chain flows. Typical examples and platforms that have been widely adopted by financial organizations are Ethereum, Hyperledger, R3 Corda, Ripple, Wave, etc. Different rationales dominate blockchain-based architectural designs, and various blockchain applications in supply chain activities demand joint collaborations among participating actors [124].

The findings we reviewed show that blockchain system planning and design is scenario-oriented [125]. We may take international trade supply chain as an example to shed light on the selection criteria for blockchain type and platform. When oriented toward disclosing information to the general public, a public chain with permissionless accessibility is preferable to improve the visibility of front-end workflow. Ethereum provides public chain settings for supply chain customers to trace physical asset provenance, authenticity and logistics status [126]. Conversely, local trade dealing with internal processes, services, and transactions may require the adoption of a permissioned private chain. Hyperledger may deploy at enterprise private network for sensitive information sharing and exchange in trade context. In an international trade setting, a consortium chain deployed between public and private chains could leverage its mediating role to facilitate transaction validation and the governance structure of grouped stakeholders [37]. The dominant players in a consortium chain play predefined roles, and its semi-decentralized structure may facilitate business operations in preferredaccess control and authority administration [127]. While R3 Corda, as a platform targeting financial applications, may support trade finance and with capability to administer agreements and contracts, Ripple focus on facilitating monetary and payment transactions among trade counterparties. Wave's pilot demonstrates blockchain's potential uses in electronic document submission which greatly result in huge reduction on transaction cost and document transmission. The aforementioned blockchain platforms are applicable to different supply chain use cases, and researchers may also conduct

comparisons of industrial pilots to understand how their features help achieve supply chain objectives [128].

The formation of a common framework may orient the level of participation since the greatest concern for supply chain players is how the framework distributes value to each entity and what value does the business capture within the blockchain ecosystem. Interactions among stakeholders and the sustainability of blockchain networks are essential for future blockchain development [129]. For example, a consortium blockchain framework may predominantly allocate governance to a specific group of leading enterprises [130]. These players act as value distributors and may orient the operations of permission control, credit evaluation, and regulation design. Participating entities may comprise suppliers, distributors, public sectors, and financial service providers. A sound framework with an efficient value-distributed ecosystem can extend both the technical and social influence of a blockchain framework.

E. THEORETICAL CONTRIBUTIONS AND RESEARCH IMPLICATIONS

In this literature review, we reexamined recent blockchain studies in the SCM context. Several studies focus on the general fact that blockchain may disrupt traditional supply chain practices. Many studies adopted the perspective of transaction cost [131]. Blockchain, as a shared distributed ledger, has the potential to mitigate trust issues among stakeholders, which in turn could alleviate business frictions, improve the visibility of supply chain activities, as well as the significant manipulations of human- and paper-based administration. With blockchain's digital ledger, participants can enjoy the timeliness of information transmission among entities. In this regard, enterprises can harvest this benefit by reducing the costs of information-related searching, bargaining, and policing [130]. Additionally, smart contracts can enhance the level of automation by facilitating the operation and execution of business processes, as seen in the system designs of the reviewed articles. Based on the principal agent theory, the use of smart contracts may improve the interactions and relations among supply chain participants, i.e., the principals and the agents in various value-chain-related industries. A collaborative framework, based on assorted blockchain platforms, can mitigate issues of information asymmetry through better control of contract execution and supervision. Blockchainbased business models in the supply chain ecosystem have been less discussed in the extant literature [132], while the need for holistic evaluation from supply chain practitioners remains strong. This phenomenon may be because blockchain development is still in a nascent stage and certain technical challenges remain unresolved. Based on the findings we reviewed, researchers have focused more on the improvement of disparate processes and business performance than the overall examinations of how blockchain adopters may derive economic gains from value creation and revenue strategies. Future research on managerial issues should address the perspective of resource allocation and

network collaboration. This theoretical work may help analyze, elaborate, and forecast the impacts of blockchain in supply chain management.

This paper has identified key research topics and the methodologies required to explore blockchain in the supply chain field. However, operational challenges such as how to determine whether blockchain adoption is required remain at the discretion of managers and practitioners. The use of blockchain should be carefully considered in terms of product (or service) characteristics, alternative surrounding technologies, and a firm's supply chain strategies. As the challenges of migrating from a legacy system to blockchain-based applications may influence the adopter's incentives, a careful consideration of implementation costs, technical limitations, barriers to entry, economic benefits, difficulties across multitier actors, and regulative compliance should be conducted before a final decision is made. In this study, Table 2 reports blockchain applications in potential industries while Table 3 summarizes the topics and methodologies adopted by researchers. The positioning of this research with its adopted methodologies contributes to the understanding of blockchain by providing an overview of blockchain research and highlighting the distribution of and gaps in extant literature. Only a few extant works of research have been devoted to large-scale adoption surveys from the perspectives of cost analysis, tradeoff considerations, implementation difficulties, and relational impacts with supply chain actors. Work on blockchain's impact on multi-tier networks and the diversity of implementation strategies in various business sectors is still scarce. This presents promising areas for future research. Academic researchers can devote their efforts on less-explored fields, while practitioners could conduct empirical studies to evaluate the effectiveness of blockchain-supply-chain adoption or implementation [133], [134]. Based on previous review studies, this work synthesizes and analyzes research contributions from recent published literature to highlight that blockchain can reform the traditional paradigm of supply chain operations and develop a more sustainable environment for the overall supply chain ecosystem [129], [135]. Finally, the call for supply chain incumbents to deliver blockchain promise based on its disruptive features may further harvest tangible benefits. More research efforts focusing on empirical evidence when adoption, technological usability, and solutions to business long-term uncertainties, are required to fill research gaps and guide managerial practices.

VI. CONCLUSION AND FUTURE WORK

This article aimed to provide a systematic review and analysis of extant literature focusing on SCM from a blockchain and smart contract perspective. Efforts to use these emerging technologies to innovate supply chain applications have fragmented research into focusing on various topics. Research efforts are evolving from technological studies, prototype architecture design, to a rather diverse focus on industrial applications, managerial implications, and social impact. To date, we are conscious of the research motivation of not only the pursuit of specific supply chain objectives but also for more open collaborative networks, autonomy, and distributed governance. The main contribution of this study is a systematic review of extant studies and citation analysis to provide a better understanding about the trajectories and applications of such technologies.

Nevertheless, future work must address several blockchainrelated technical issues such as throughput, security, scalability, and interoperability. Similar efforts are limited and related quantitative study regarding these topics is still rare. More research work needs to address the diffusion of blockchain technology, BPR, and their managerial implications and social impact. Promising areas for blockchain applications and research include the logistics, medical, insurance, and public sectors. We expect this study to provide an overview of extant blockchain-related supply chain research and shed light on potential research gaps and directions for future works.

REFERENCES

- R. Cole, M. Stevenson, and J. Aitken, "Blockchain technology: Implications for operations and supply chain management," *Supply Chain Manag.*, vol. 24, no. 4, pp. 469–483, Jun. 2019.
- [2] F. Hawlitschek, B. Notheisen, and T. Teubner, "The limits of trust-free systems: A literature review on blockchain technology and trust in the sharing economy," *Electron. Commerce Res. Appl.*, vol. 29, pp. 50–63, May 2018.
- [3] J. Mendling, "Blockchains for business process management challenges and opportunities," ACM Trans. Manag. Inform. Syst., vol. 9, no. 1, p. 4, 2018.
- [4] X. Pan, X. Pan, M. Song, B. Ai, and Y. Ming, "Blockchain technology and enterprise operational capabilities: An empirical test," *Int. J. Inf. Manage.*, vol. 52, Jun. 2020, Art. no. 101946.
- [5] Y. Guo and C. Liang, "Blockchain application and outlook in the banking industry," *Financial Innov.*, vol. 2, no. 1, p. 24, Dec. 2016.
- [6] A. Azaria, A. Ekblaw, T. Vieira, and A. Lippman, "MedRec: Using blockchain for medical data access and permission management," in *Proc. 2nd Int. Conf. Open Big Data (OBD)*, Aug. 2016, pp. 25–30.
- [7] M. Pawlak, A. Poniszewska-Marańda, and N. Kryvinska, "Towards the intelligent agents for blockchain e-voting system," *Procedia Comput. Sci.*, vol. 141, pp. 239–246, Jan. 2018.
- [8] A. Sheth and H. Subramanian, "Blockchain and contract theory: Modeling smart contracts using insurance markets," *Managerial Finance*, May 2019, doi: 10.1108/MF-10-2018-0510.
- [9] J. Al-Jaroodi and N. Mohamed, "Blockchain in industries: A survey," *IEEE Access*, vol. 7, pp. 36500–36515, 2019.
- [10] Y. Wang, M. Singgih, J. Wang, and M. Rit, "Making sense of blockchain technology: How will it transform supply chains?" *Int. J. Prod. Econ.*, vol. 211, pp. 221–236, May 2019.
- [11] M. Pilkington, "Blockchain technology: Principles and applications," in *Research Handbook on Digital Transformations*, F. X. Olleros and M. Zhegu, Eds. Cheltenham, U.K.: Edward Elgar, 2016, pp. 2252–2253.
- [12] M. Casey and P. Wong, "Global supply chains are about to get better, thanks to blockchain," *Harv. Bus. Rev.*, vol. 13, pp. 1–6, 2017. Accessed: Sep. 1, 2018. [Online]. Available: https://hbr.org/2017/03/global-supplychains-are-about-to-get-better-thanks-to-blockchain
- [13] Z. Zheng, S. Xie, H. N. Dai, X. Chen, and H. Wang, "Blockchain challenges and opportunities: A survey," *Int. J. Web Grid Services*, vol. 14, no. 4, pp. 352–375, 2018, doi: 10.1504/IJWGS.2018.095647.
- [14] F. Tschorsch and B. Scheuermann, "Bitcoin and beyond: A technical survey on decentralized digital currencies," *IEEE Commun. Surveys Tuts.*, vol. 18, no. 3, pp. 2084–2123, 3rd Quart., 2016.
- [15] X. Xu, C. Pautasso, L. Zhu, V. Gramoli, A. Ponomarev, A. B. Tran, and S. Chen, "The blockchain as a software connector," in *Proc. 13th Work. IEEE/IFIP Conf. Softw. Archit. (WICSA)*, Venice, Italy, Apr. 2016, pp. 182–191.

- [17] E. A. Morse, "From rai stones to blockchains: The transformation of payments," *Comput. Law Secur. Rev.*, vol. 34, no. 4, pp. 946–953, Aug. 2018.
- [18] F. Sander, J. Semeijn, and D. Mahr, "The acceptance of blockchain technology in meat traceability and transparency," *Brit. Food J.*, vol. 120, no. 9, pp. 2066–2079, Sep. 2018.
- [19] Y. Wang, J. H. Han, and P. Beynon-Davies, "Understanding blockchain technology for future supply chains: A systematic literature review and research agenda," *Supply Chain Manage.*, *Int. J.*, vol. 24, no. 1, pp. 62–84, Jan. 2019.
- [20] M. M. Queiroz, R. Telles, and S. H. Bonilla, "Blockchain and supply chain management integration: A systematic review of the literature," *Supply Chain Manage.*, vol. 25, no. 2, pp. 241–254, Aug. 2019.
- [21] W. Viriyasitavat and D. Hoonsopon, "Blockchain characteristics and consensus in modern business processes," J. Ind. Inf. Integr., vol. 13, pp. 32–39, Mar. 2019.
- [22] I. Weber, X. Xu, R. Riveret, G. Governatori, A. Ponomarev, and J. Mendling, "Untrusted business process monitoring and execution using blockchain," *Lect. Notes Comput. Sci.*, vol. 9850, pp. 329–347, Sep. 2016.
- [23] S. Nakamoto. Bitcoin: A Peer-to-Peer Electronic Cash System. Accessed: Dec. 1, 2018. [Online]. Available: https://bitcoin.org/bitcoin.pdf
- [24] M. Crosby, P. Pattanayak, S. Verma, and V. Kalyanaraman, "Blockchain technology: Beyond bitcoin," *Appl. Innov.*, vol. 2, nos. 6–10, p. 71, 2016.
- [25] R. Neisse, G. Steri, and I. Nai-Fovino, "A blockchain-based approach for data accountability and provenance tracking," in *Proc. 12th Int. Conf. Availability, Rel. Secur. (ARES)*, Reggio Calabria, Italy, 2017, p. 14.
- [26] G. R. T. White, "Future applications of blockchain in business and management: A delphi study," *Strategic Change*, vol. 26, no. 5, pp. 439–451, Sep. 2017.
- [27] K. Toyoda, P. T. Mathiopoulos, I. Sasase, and T. Ohtsuki, "A novel blockchain-based product ownership management system (POMS) for anti-counterfeits in the post supply chain," *IEEE Access*, vol. 5, pp. 17465–17477, 2017.
- [28] N. Szabo, (1996). Smart Contracts: Building Blocks for Digital Markets, Accessed: Dec. 1, 2018. [Online]. Available: http://www.fon.hum.uva.nl/ rob/Courses/InformationInSpeech/CDROM/Literature/LOTwinterschool 2006/szabo.best.vwh.net/smart_contracts_2.html
- [29] D. Magazzeni, P. McBurney, and W. Nash, "Validation and verification of smart contracts: A research agenda," *Computer*, vol. 50, no. 9, pp. 50–57, 2017.
- [30] H. M. Kim and M. Laskowski, "Toward an ontology-driven blockchain design for supply chain provenance," *Intell. Syst. Account. Financ. Manag.*, vol. 25, no. 1, pp. 18–27, Mar. 2018.
- [31] H. Watanabe, S. Fujimura, A. Nakadaira, Y. Miyazaki, A. Akutsu, and J. J. Kishigami, "Blockchain contract: A complete consensus using blockchain," in *Proc. IEEE 4th Global Conf. Consum. Electron. (GCCE)*, Osaka, Japan, Oct. 2015, pp. 577–578.
- [32] C. Sturm, J. Scalanczi, S. Schönig, and S. Jablonski, "A blockchain-based and resource-aware process execution engine," *Future Gener. Comput. Syst.*, vol. 100, pp. 19–34, Nov. 2019.
- [33] Y. Wang and A. Kogan, "Designing confidentiality-preserving blockchain-based transaction processing systems," *Int. J. Accounting Inf. Syst.*, vol. 30, pp. 1–18, Sep. 2018.
- [34] D. E. O'Leary, "Configuring blockchain architectures for transaction information in blockchain," *Intell. Syst. Account. Finance Manag.*, vol. 24, no. 4, pp. 138–147, Dec. 2017.
- [35] V. Buterin, (2014), A Next-Generation Smart Contract and Decentralized Application Platform. Accessed: Sep. 1, 2019. [Online]. Available: https://cryptorating.eu/whitepapers/Ethereum/Ethereum_white_ paper.pdf
- [36] A. Dolgui, D. Ivanov, S. Potryasaev, B. Sokolov, M. Ivanova, and F. Werner, "Blockchain-oriented dynamic modelling of smart contract design and execution in the supply chain," *Int. J. Prod. Res.*, vol. 57, pp. 1–16, Jun. 2019.
- [37] S. E. Chang, Y.-C. Chen, and T.-C. Wu, "Exploring blockchain technology in international trade: Business process re-engineering for letter of credit," *Ind. Manage. Data Syst.*, vol. 119, no. 8, pp. 1712–1733, Aug. 2019.

- [38] S. Wang, L. Ouyang, Y. Yuan, X. Ni, X. Han, and F.-Y. Wang, "Blockchain-enabled smart contracts: Architecture, applications, and future trends," *IEEE Trans. Syst., Man, Cybern. Syst.*, vol. 49, no. 11, pp. 2266–2277, Nov. 2019.
- [39] K. Delmolino, M. Arnett, A. Kosba, A. Miller, and E. Shi, "Step by step towards creating a safe smart contract: Lessons and insights from a cryptocurrency lab," in *Proc. Int. Conf. Financ. Cryptogr. Data Secur.*, Berline, German, Aug. 2016, pp. 79–94.
- [40] C. Prybila, S. Schulte, C. Hochreiner, and I. Weber, "Runtime verification for business processes utilizing the Bitcoin blockchain," *Future Gener. Comput. Syst.*, vol. 107, pp. 816–831, Jun. 2020.
- [41] A. Kosba, A. Miller, E. Shi, Z. Wen, and C. Papamanthou, "Hawk: The blockchain model of cryptography and privacy-preserving smart contracts," in *Proc. IEEE Symp. Secur. Privacy (SP)*, San Jose, CA, USA, May 2016, pp. 839–858.
- [42] F.-Y. Wang, Y. Yuan, C. Rong, and J. J. Zhang, "Parallel blockchain: An architecture for CPSS-based smart societies," *IEEE Trans. Comput. Social Syst.*, vol. 5, no. 2, pp. 303–310, Jun. 2018.
- [43] Y. Zhang, S. Kasahara, Y. Shen, X. Jiang, and J. Wan, "Smart contract-based access control for the Internet of Things," *IEEE Internet Things J.*, vol. 6, no. 2, pp. 1594–1605, Apr. 2019.
- [44] D. Tranfield, D. Denyer, and P. Smart, "Towards a methodology for developing evidence-informed management knowledge by means of systematic review," *Brit. J. Manage.*, vol. 14, no. 3, pp. 207–222, Sep. 2003.
- [45] B. Kitchenham and P. Brereton, "A systematic review of systematic review process research in software engineering," *Inf. Softw. Technol.*, vol. 55, no. 12, pp. 2049–2075, Dec. 2013.
- [46] H. Walker and A. Touboulic, "Theories in sustainable supply chain management: A structured literature review," *Int. J. Phys. Distrib. Logist. Manag.*, vol. 45, nos. 1–2, pp. 16–42, Mar. 2015.
- [47] R. Azzi, R. K. Chamoun, and M. Sokhn, "The power of a blockchainbased supply chain," *Comput. Ind. Eng.*, vol. 135, pp. 582–592, Sep. 2019.
- [48] G. Perboli, S. Musso, and M. Rosano, "Blockchain in logistics and supply chain: A lean approach for designing real-world use cases," *IEEE Access*, vol. 6, pp. 62018–62028, 2018.
- [49] S. Figorilli, F. Antonucci, C. Costa, F. Pallottino, L. Raso, M. Castiglione, E. Pinci, D. Del Vecchio, G. Colle, A. Proto, G. Sperandio, and P. Menesatti, "A blockchain implementation prototype for the electronic open source traceability of wood along the whole supply chain," *Sensors*, vol. 18, no. 9, p. 3133, 2018.
- [50] R. V. George, H. O. Harsh, P. Ray, and A. K. Babu, "Food quality traceability prototype for restaurants using blockchain and food quality data index," *J. Clean. Prod.*, vol. 240, Dec. 2019, Art. no. 118021.
- [51] Q. Lin, H. Wang, X. Pei, and J. Wang, "Food safety traceability system based on blockchain and EPCIS," *IEEE Access*, vol. 7, pp. 20698–20707, 2019.
- [52] K. Behnke and M. F. W. H. A. Janssen, "Boundary conditions for traceability in food supply chains using blockchain technology," *Int. J. Inf. Manage.*, vol. 52, Jun. 2020, Art. no. 101969.
- [53] S. Bose, M. Raikwar, D. Mukhopadhyay, A. Chattopadhyay, and K.-Y. Lam, "BLIC: A blockchain protocol for manufacturing and supply chain management of ICS," in *Proc. IEEE Int. Conf. Internet Things (iThings) IEEE Green Comput. Commun. (GreenCom) IEEE Cyber, Phys. Social Comput. (CPSCom) IEEE Smart Data (SmartData)*, Jul. 2018, pp. 1326–1335.
- [54] M. Westerkamp, F. Victor, and A. Kupper, "Blockchain-based supply chain traceability: Token recipes model manufacturing processes," in Proc. IEEE Int. Conf. Internet Things (iThings) IEEE Green Comput. Commun. (GreenCom) IEEE Cyber, Phys. Social Comput. (CPSCom) IEEE Smart Data (SmartData), Jul. 2018, pp. 1595–1602.
- [55] T. Bocek, B. B. Rodrigues, T. Strasser, and B. Stiller, "Blockchains everywhere-a use-case of blockchains in the pharma supply-chain," in *Proc. IFIP/IEEE Symp. Integr. Netw. Service Manage. (IM)*, May 2017, pp. 772–777.
- [56] Y. Huang, J. Wu, and C. Long, "Drugledger: A practical blockchain system for drug traceability and regulation," in *Proc. IEEE Int. Conf. Internet Things (iThings) IEEE Green Comput. Commun. (GreenCom) IEEE Cyber, Phys. Social Comput. (CPSCom) IEEE Smart Data (Smart-Data)*, Jul. 2018, pp. 1137–1144.
- [57] F. Jamil, L. Hang, K. Kim, and D. Kim, "A novel medical blockchain model for drug supply chain integrity management in a smart hospital," *Electronics*, vol. 8, no. 5, p. 505, 2019.

- [58] Q. Lu and X. Xu, "Adaptable blockchain-based systems: A case study for product traceability," *IEEE Softw.*, vol. 34, no. 6, pp. 21–27, Nov. 2017.
- [59] J. Astill, R. A. Dara, M. Campbell, J. M. Farber, E. D. G. Fraser, S. Sharif, and R. Y. Yada, "Transparency in food supply chains: A review of enabling technology solutions," *Trends Food Sci. Technol.*, vol. 91, pp. 240–247, Sep. 2019.
- [60] J. J. Sikorski, J. Haughton, and M. Kraft, "Blockchain technology in the chemical industry: Machine-to-machine electricity market," *Appl. Energy*, vol. 195, pp. 234–246, Jun. 2017.
- [61] J. Wang, P. Wu, X. Wang, and W. Shou, "The outlook of blockchain technology for construction engineering management," *Frontiers Eng. Manage.*, vol. 4, no. 1, pp. 67–75, 2017.
- [62] F. Xiong, R. Xiao, W. Ren, R. Zheng, and J. Jiang, "A key protection scheme based on secret sharing for blockchain-based construction supply chain system," *IEEE Access*, vol. 7, pp. 126773–126786, 2019.
- [63] W. Ying, S. Jia, and W. Du, "Digital enablement of blockchain: Evidence from HNA group," *Int. J. Inf. Manage.*, vol. 39, pp. 1–4, Apr. 2018.
- [64] C. Mandolla, A. M. Petruzzelli, G. Percoco, and A. Urbinati, "Building a digital twin for additive manufacturing through the exploitation of blockchain: A case analysis of the aircraft industry," *Comput. Ind.*, vol. 109, pp. 134–152, Aug. 2019.
- [65] T.-M. Choi, X. Wen, X. Sun, and S.-H. Chung, "The mean-variance approach for global supply chain risk analysis with air logistics in the blockchain technology era," *Transp. Res. E, Logistics Transp. Rev.*, vol. 127, pp. 178–191, Jul. 2019.
- [66] Z. Li, A. V. Barenji, and G. Q. Huang, "Toward a blockchain cloud manufacturing system as a peer to peer distributed network platform," *Robot. Comput.-Integr. Manuf.*, vol. 54, pp. 133–144, Dec. 2018.
- [67] M. N. Islam and S. Kundu, "Enabling IC traceability via blockchain pegged to embedded PUF," ACM Trans. Design Autom. Electron. Syst., vol. 24, no. 3, pp. 1–23, Apr. 2019.
- [68] A. Reyna, C. Martín, J. Chen, E. Soler, and M. Díaz, "On blockchain and its integration with IoT. Challenges and opportunities," *Future Gener. Comput. Syst.*, vol. 88, pp. 173–190, Nov. 2018.
- [69] S. Mondal, K. P. Wijewardena, S. Karuppuswami, N. Kriti, D. Kumar, and P. Chahal, "Blockchain inspired RFID-based information architecture for food supply chain," *IEEE Internet Things J.*, vol. 6, no. 3, pp. 5803–5813, Jun. 2019.
- [70] T. M. Fernández-Caramés, O. Blanco-Novoa, I. Froiz-Míguez, and P. Fraga-Lamas, "Towards an autonomous industry 4.0 warehouse: A UAV and blockchain-based system for inventory and traceability applications in big data-driven supply chain management," *Sensors*, vol. 19, no. 10, p. 2394, 2019.
- [71] M. Sidorov, M. T. Ong, R. V. Sridharan, J. Nakamura, R. Ohmura, and J. H. Khor, "Ultralightweight mutual authentication RFID protocol for blockchain enabled supply chains," *IEEE Access*, vol. 7, pp. 7273–7285, 2019.
- [72] E. Tijan, S. Aksentijević, K. Ivanić, and M. Jardas, "Blockchain technology implementation in logistics," *Sustainability*, vol. 11, no. 4, p. 1185, 2019.
- [73] S. Tönnissen and F. Teuteberg, "Analysing the impact of blockchaintechnology for operations and supply chain management: An explanatory model drawn from multiple case studies," *Int. J. Inf. Manage.*, vol. 52, Jun. 2020, Art. no. 101953.
- [74] M. Li, L. Shen, and G. Q. Huang, "Blockchain-enabled workflow operating system for logistics resources sharing in E-commerce logistics real estate service," *Comput. Ind. Eng.*, vol. 135, pp. 950–969, Sep. 2019.
- [75] L. Liu, F. Li, and E. Qi, "Research on risk avoidance and coordination of supply chain subject based on blockchain technology," *Sustainability*, vol. 11, no. 7, p. 2182, 2019.
- [76] Y. Fu and J. Zhu, "Big production enterprise supply chain endogenous risk management based on blockchain," *IEEE Access*, vol. 7, pp. 15310–15319, 2019.
- [77] S. Kamble, A. Gunasekaran, and H. Arha, "Understanding the blockchain technology adoption in supply chains-indian context," *Int. J. Prod. Res.*, vol. 57, no. 7, pp. 2009–2033, Apr. 2019.
- [78] L.-W. Wong, L.-Y. Leong, J.-J. Hew, G. W.-H. Tan, and K.-B. Ooi, "Time to seize the digital evolution: Adoption of blockchain in operations and supply chain management among Malaysian SMEs," *Int. J. Inf. Manage.*, vol. 52, Jun. 2020, Art. no. 101997.
- [79] M. M. Queiroz and S. Fosso Wamba, "Blockchain adoption challenges in supply chain: An empirical investigation of the main drivers in India and the USA," *Int. J. Inf. Manage.*, vol. 46, pp. 70–82, Jun. 2019.

- [80] A. Gausdal, K. Czachorowski, and M. Solesvik, "Applying blockchain technology: Evidence from Norwegian companies," *Sustainability*, vol. 10, no. 6, p. 1985, 2018.
- [81] N. Hackius and M. Petersen, "Blockchain in logistics and supply chain: Trick or treat?" in *Proc. Hamburg Int. Conf. Logist. (HICL)*, Hamburg, German, vol. 23y, Jul. 2017, pp. 3–18.
- [82] H. Wu, Z. Li, B. King, Z. Ben Miled, J. Wassick, and J. Tazelaar, "A distributed ledger for supply chain physical distribution visibility," *Information*, vol. 8, no. 4, p. 137, 2017.
- [83] H. Kim and M. Laskowski, "A perspective on blockchain smart contracts: Reducing uncertainty and complexity in value exchange," in *Proc. 26th Int. Conf. Comput. Commun. Netw. (ICCCN)*, Vancouver, BC, Canada, Jul./Aug. 2017, pp. 1–6.
- [84] F. Tian, "An agri-food supply chain traceability system for China based on RFID & blockchain technology," in *Proc. 13th Int. Conf. Service Syst. Service Manag. (ICSSSM)*, Kunming, China, Jun. 2016, pp. 1–6.
- [85] F. Tian, "A supply chain traceability system for food safety based on HACCP, blockchain and Internet of Things," in *Proc. Int. Conf. Service Syst. Service Manag.*, Dalian, China, Jun. 2017, pp. 1–6.
- [86] K. Leng, Y. Bi, L. Jing, H.-C. Fu, and I. Van Nieuwenhuyse, "Research on agricultural supply chain system with double chain architecture based on blockchain technology," *Future Gener. Comput. Syst.*, vol. 86, pp. 641–649, Sep. 2018.
- [87] K. Salah, N. Nizamuddin, R. Jayaraman, and M. Omar, "Blockchainbased soybean traceability in agricultural supply chain," *IEEE Access*, vol. 7, pp. 73295–73305, 2019.
- [88] S. S. Kamble, A. Gunasekaran, and R. Sharma, "Modeling the blockchain enabled traceability in agriculture supply chain," *Int. J. Inf. Manage.*, vol. 52, Jun. 2020, Art. no. 101967.
- [89] F. M. Bencic, P. Skocir, and I. P. Zarko, "DL-tags: DLT and smart tags for decentralized, privacy-preserving, and verifiable supply chain management," *IEEE Access*, vol. 7, pp. 46198–46209, 2019.
- [90] Q. Feng, D. He, S. Zeadally, M. K. Khan, and N. Kumar, "A survey on privacy protection in blockchain system," *J. Netw. Comput. Appl.*, vol. 126, pp. 45–58, Jan. 2019.
- [91] Z. Gao, L. Xu, L. Chen, X. Zhao, Y. Lu, and W. Shi, "CoC: A unified distributed ledger based supply chain management system," *J. Comput. Sci. Technol.*, vol. 33, no. 2, pp. 237–248, Mar. 2018.
- [92] N. Kshetri, "Blockchain's roles in strengthening cybersecurity and protecting privacy," *Telecommun. Policy*, vol. 41, no. 10, pp. 1027–1038, Nov. 2017.
- [93] S. Rouhani and R. Deters, "Security, performance, and applications of smart contracts: A systematic survey," *IEEE Access*, vol. 7, pp. 50759–50779, 2019.
- [94] D. Mao, F. Wang, Z. Hao, and H. Li, "Credit evaluation system based on blockchain for multiple stakeholders in the food supply chain," *Int. J. Environ. Res. Public Health*, vol. 15, no. 8, p. 1627, 2018.
- [95] V. Shermin, "Disrupting governance with blockchains and smart contracts," *Strategic Change*, vol. 26, no. 5, pp. 499–509, Sep. 2017.
- [96] J.-H. Tseng, Y.-C. Liao, B. Chong, and S.-W. Liao, "Governance on the drug supply chain via gcoin blockchain," *Int. J. Environ. Res. Public Health*, vol. 15, no. 6, p. 1055, 2018.
- [97] N. Kshetri, "1 Blockchain's roles in meeting key supply chain management objectives," *Int. J. Inf. Manage.*, vol. 39, pp. 80–89, Apr. 2018.
- [98] F. Casino, T. K. Dasaklis, and C. Patsakis, "A systematic literature review of blockchain-based applications: Current status, classification and open issues," *Telematics Informat.*, vol. 36, pp. 55–81, Mar. 2019.
- [99] L. Hughes, Y. K. Dwivedi, S. K. Misra, N. P. Rana, V. Raghavan, and V. Akella, "Blockchain research, practice and policy: Applications, benefits, limitations, emerging research themes and research agenda," *Int. J. Inf. Manage.*, vol. 49, pp. 114–129, Dec. 2019.
- [100] T. K. Mackey, T.-T. Kuo, B. Gummadi, K. A. Clauson, G. Church, D. Grishin, K. Obbad, R. Barkovich, and M. Palombini, "'Fit-forpurpose?'—Challenges and opportunities for applications of blockchain technology in the future of healthcare," *BMC Med.*, vol. 17, no. 1, p. 68, Dec. 2019.
- [101] M. Pournader, Y. Y. Shi, S. Seuring, and S. C. L. Koh, "Blockchain applications in supply chains, transport and logistics: A systematic review of the literature," *Int. J. Prod. Res.*, vol. 57, pp. 1–19, Aug. 2019.
- VOLUME 8, 2020

- [102] E. B. Sifah, A. Shahzad, S. Amofa, S. Abla, K. O.-B. O. Agyekum, Q. Xia, and J. Gao, "Blockchain based monitoring on trustless supply chain processes," in *Proc. IEEE Int. Conf. Internet Things (iThings) IEEE Green Comput. Commun. (GreenCom) IEEE Cyber Phys. Social Comput. (CPSCom) IEEE Smart Data (SmartData)*, Jul./Aug. 2018, pp. 1189–1195.
- [103] M. Yoo and Y. Won, "A study on the transparent price tracing system in supply chain management based on blockchain," *Sustainability*, vol. 10, no. 11, p. 4037, 2018.
- [104] H. Hasan, E. AlHadhrami, A. AlDhaheri, K. Salah, and R. Jayaraman, "Smart contract-based approach for efficient shipment management," *Comput. Ind. Eng.*, vol. 136, pp. 149–159, Oct. 2019.
- [105] P. Helo and Y. Hao, "Blockchains in operations and supply chains: A model and reference implementation," *Comput. Ind. Eng.*, vol. 136, pp. 242–251, Oct. 2019.
- [106] S. S. Kamble, A. Gunasekaran, and S. A. Gawankar, "Achieving sustainable performance in a data-driven agriculture supply chain: A review for research and applications," *Int. J. Prod. Econ.*, vol. 219, pp. 179–194, Jan. 2020.
- [107] K. S. Hald and A. Kinra, "How the blockchain enables and constrains supply chain performance," *Int. J. Phys. Distrib. Logistics Manage.*, vol. 49, no. 4, pp. 376–397, Jun. 2019.
- [108] P. K. Sharma, N. Kumar, and J. H. Park, "Blockchain-based distributed framework for automotive industry in a smart city," *IEEE Trans. Ind. Informat.*, vol. 15, no. 7, pp. 4197–4205, Jul. 2019.
- [109] S. Pearson, D. May, G. Leontidis, M. Swainson, S. Brewer, L. Bidaut, J. G. Frey, G. Parr, R. Maull, and A. Zisman, "Are distributed ledger technologies the panacea for food traceability?" *Global Food Secur.*, vol. 20, pp. 145–149, Mar. 2019.
- [110] J. F. Galvez, J. C. Mejuto, and J. Simal-Gandara, "Future challenges on the use of blockchain for food traceability analysis," *TrAC Trends Anal. Chem.*, vol. 107, pp. 222–232, Oct. 2018.
- [111] Q. Qu, I. Nurgaliev, M. Muzammal, C. S. Jensen, and J. Fan, "On spatiotemporal blockchain Query processing," *Future Gener. Comput. Syst.*, vol. 98, pp. 208–218, Sep. 2019.
- [112] S. Yadav and S. P. Singh, "Blockchain critical success factors for sustainable supply chain," *Resour, Conservation Recycling*, vol. 152, Jan. 2020, Art. no. 104505.
- [113] M. Nakasumi, "Information sharing for supply chain management based on block chain technology," in *Proc. IEEE 19th Conf. Bus. Informat.* (*CBI*), Thessaloniki, Greece, Jul. 2017, pp. 140–149.
- [114] S. V. Engelenburg, M. Janssen, and B. Klievink, "Design of a software architecture supporting business-to-government information sharing to improve public safety and security," *J. Intell. Inf. Syst.*, vol. 52, no. 3, pp. 595–618, Jun. 2019.
- [115] K. Korpela, J. Hallikas, and T. Dahlberg, "Digital supply chain transformation toward blockchain integration," in *Proc. Hawaii Int. Conf. Syst. Sci.*, Jan. 2017, pp. 4182–4191.
- [116] A. Imeri and D. Khadraoui, "The security and traceability of shared information in the process of transportation of dangerous goods," in *Proc.* 9th IFIP Int. Conf. New Technol., Mobility Secur. (NTMS), Feb. 2018, pp. 1–5.
- [117] W. Nowiński and M. Kozma, "How can blockchain technology disrupt the existing business models?" *Entrepreneurial Bus. Econ. Rev.*, vol. 5, no. 3, pp. 173–188, Jun. 2017.
- [118] S. Rahmanzadeh, M. S. Pishvaee, and M. R. Rasouli, "Integrated innovative product design and supply chain tactical planning within a blockchain platform," *Int. J. Prod. Res.*, vol. 57, pp. 1–21, Aug. 2019.
- [119] C. G. Schmidt and S. M. Wagner, "Blockchain and supply chain relations: A transaction cost theory perspective," *J. Purchasing Supply Manage.*, vol. 25, no. 4, Oct. 2019, Art. no. 100552.
- [120] X. Xu, F. Rahman, B. Shakya, A. Vassilev, D. Forte, and M. Tehranipoor, "Electronics supply chain integrity enabled by blockchain," *ACM Trans. Design Autom. Electron. Syst.*, vol. 24, no. 3, pp. 1–25, May 2019.
- [121] M. Swan, Blockchain: Blueprint for a New Economy, 1st ed. Newton, MA, USA: O'Reilly Media, 2015.
- [122] J. S. Srai and H. Lorentz, "Developing design principles for the digitalisation of purchasing and supply management," J. Purchasing Supply Manage., vol. 25, no. 1, pp. 78–98, Jan. 2019.
- [123] Z. Li, H. Wu, B. King, Z. B. Miled, J. Wassick, and J. Tazelaar, "On the integration of event-based and transaction-based architectures for supply chains," in *Proc. IEEE 37th Int. Conf. Distrib. Comput. Syst. Workshops (ICDCSW)*, Jun. 2017, pp. 376–382.

- [124] X. Xu, Q. Lu, Y. Liu, L. Zhu, H. Yao, and A. V. Vasilakos, "Designing blockchain-based applications a case study for imported product traceability," *Future Gener. Comput. Syst.*, vol. 92, pp. 399–406, Mar. 2019.
- [125] S. Wang, D. Li, Y. Zhang, and J. Chen, "Smart contract-based product traceability system in the supply chain scenario," *IEEE Access*, vol. 7, pp. 115122–115133, 2019.
- [126] F. Longo, L. Nicoletti, A. Padovano, G. d'Atri, and M. Forte, "Blockchain-enabled supply chain: An experimental study," *Comput. Ind. Eng.*, vol. 136, pp. 57–69, Oct. 2019.
- [127] R. Qiao, S. Zhu, Q. Wang, and J. Qin, "Optimization of dynamic data traceability mechanism in Internet of Things based on consortium blockchain," *Int. J. Distrib. Sensor Netw.*, vol. 14, no. 12, Dec. 2018, Art. no. 155014771881907.
- [128] R. van Hoek, "Developing a framework for considering blockchain pilots in the supply chain - lessons from early industry adopters," *Supply Chain Manag.*, vol. 25, no. 1, pp. 115–121, 2020.
- [129] S. Saberi, M. Kouhizadeh, J. Sarkis, and L. Shen, "Blockchain technology and its relationships to sustainable supply chain management," *Int. J. Prod. Res.*, vol. 57, no. 7, pp. 2117–2135, Apr. 2019.
- [130] S. E. Chang, H. L. Luo, and Y.-C. Chen, "Blockchain-enabled trade finance innovation: A potential paradigm shift on using letter of credit," *Sustainability*, vol. 12, no. 1, p. 188, Jan. 2020.
- [131] H. Treiblmaier, "The impact of the blockchain on the supply chain: A theory-based research framework and a call for action," *Supply Chain Manage.*, vol. 23, no. 6, pp. 545–559, Sep. 2018.
- [132] C. Bai and J. Sarkis, "A supply chain transparency and sustainability technology appraisal model for blockchain technology," *Acad. Manage. Proc.*, vol. 2019, no. 1, p. 16069, Aug. 2019.
- [133] R. van Hoek, "Unblocking the chain—Findings from an executive workshop on blockchain in the supply chain," *Supply Chain Manage.*, vol. 25, no. 2, pp. 255–261, Jun. 2019.
- [134] A. Sheel and V. Nath, "Effect of blockchain technology adoption on supply chain adaptability, agility, alignment and performance," *Manage. Res. Rev.*, vol. 42, no. 12, pp. 1353–1374, Dec. 2019.
- [135] M. Kouhizadeh and J. Sarkis, "Blockchain practices, potentials, and perspectives in greening supply chains," *Sustainability*, vol. 10, no. 10, p. 3652, Oct. 2018.



SHUCHIH E. CHANG received the M.S. degree in computer science and the Ph.D. degree from The University of Texas at Austin. He has 15 years of working experience in major computer and financial service firms in the USA, including Unisys, IBM, JP Morgan, Bear Stearns, and UBS. He is a Distinguished Professor with the Graduate Institute of Technology Management, National Chung Hsing University (NCHU), Taiwan. Before joining the Faculty at NCHU, he worked with UBS Finan-

cial Services Inc., USA, as the Divisional Vice President, for about five years. His publications have appeared in *Computers in Human Behavior*, the *Journal of Business Research, Technological Forecasting & Social Changes, Expert Systems With Applications*, the IEEE PERVASIVE COMPUTING, *Information and Software Technology*, the *Journal of Organizational Computing and Electronic Commerce*, the *International Journal of Technology Management*, *Behaviour and Information Technology, Industrial Management & Data Systems,* and *Multimedia Tools and Applications.* His research interests are in financial technology, blockchain applications, technology management, electronic commerce, enterprise information architecture, and information security management.



YICHIAN CHEN received the M.S. degree in applied mechanics and engineering from National Taiwan University, in 2011. He is currently pursuing the Ph.D. degree with the Graduate Institute of Technology Management, National Chung Hsing University, Taiwan. His research interests are in technology management, blockchain and supply chain management, and information services and its applications.

. . .